Author's response to reviews

Title: Novel SPAST deletion and reduced DPY30 expression in a Spastic Paraplegia type 4 kindred

Authors:

- Loretta Racis (racis.l@libero.it)
- Eugenia Storti (storti.e@tiscali.it)
- Maura Pugliatti (mpugliatti@yahoo.com)
- Virgilio Agnetti (agnetti@uniss.it)
- Alessandra Tessa (aletessa@gmail.com)
- Filippo M. Santorelli (filippo3364@gmail.com)

Version: 3  Date: 11 March 2014

Author's response to reviews:

Editor-in-Chief
BMC Medical Genetics
11 March 2014
RE: MS: 1434273587118720-R2

Dear Editor,

Enclosed please find the re-revised version of a manuscript entitled “Novel SPAST deletion and reduced DPY30 expression in a Spastic Paraplegia type 4 kindred” by doctors Racis, colleagues and myself, for possible publication in BMC Medical Genetics.

We have modified the text according to your suggestions. Changes have been highlighted in the re-revised manuscript.

Editor

1. Please include the email addresses of all co-authors in the title page.

The e-mail addresses of all-co-authors have been added, as requested.

We hope that in this revised version the manuscript will be suitable for publication.

Sincerely,

Filippo M. Santorelli
filippo3364@gmail.com